Author's response to reviews

Title: Acute scrotum as a complication of Thiersch operation for rectal prolapse in a child

Authors:

Mohammad M Saleem (mohomari@hotmail.com)
Hashem M Al-momani (hashemmomani@yahoo.com)

Version: 3 Date: 2 December 2006

Author's response to reviews: see over
Dear Sir

In response to all the reviewers’ comments, I hope I have answered all the questions asked and the grammar was corrected as the manuscript was read by two native English speaking physicians. I hope I have met the entire requirement for acceptance of my case report.

As a response to the reviewer (DR. Gupta)

Dear Sir

This is a point by point answer to your questions

1- The condition of the tract: as shown in the figure there was no inflammation or suppuration, but there was moderate amount of induration, and fibrosis at the anal verge.
2- Anal canal and rectum: PR examination was normal without any unusual findings to suggest paraproctitis; there was moderate amount of narrowing at the indurated anal verge.
3- Rectal prolapse: It was cured
4- Endoanal U/S: It was not done, as it was not thought to be necessary at the time.
5- Follow up: Was uneventful both to the scrotal redness and swelling as well as to the rectal prolapse.

These points were added to the text, as indicated.

Sincerely yours
Mohammad M Saleem MD